Author’s response to reviews

Title: Title: Incidence and determinants of diabetes-related lower limb amputations in Ghana, 2010-2015- a retrospective cohort study

Authors:

OSEI SARFO-KANTANKA (osarfokantanka21@gmail.com)

Fred Sarfo (stephensarfo78@gmail.com)

Ishmael Kyei (raskyei@yahoo.com)

Charles Agyemang (C.O.Agyemang@amc.uva.nl)

Jean Claude Mbanya (jcmbanya@yahoo.co.uk)

Version: 1 Date: 06 Apr 2018

Author’s response to reviews:

6th April, 2018.

The Editor,

BMC Endocrine Disorders

Dear Editor,

We are most grateful for the attention given to reviewing our manuscript. We have used the comments and suggestions by the Reviewers to strengthen the manuscript. The editorial comments have been responded to below:

In the discussion, I would like to see more of an emphasis on the risk of bias as this is a retrospective and also a single center study. Explore that issue more.
The risk of bias based on fact that the study is single center study and retrospective in design has been highlighted in the corrected manuscript, Page 11, line 211-215.

Also I would reduce the number of references to about 30.

The list of References has been reduced to 30 as requested.

Lastly the figures have had some "track changes" comments retained, these need to be removed.

Tracked changes on the any of the tables or figures has been removed.

Sincerely,

Dr Osei Sarfo-Kantanka